Author’s response to reviews

Title: Impact of Rapid Investigation Clinic on timeliness of lung cancer diagnosis and treatment

Authors:

Nicole Ezer (Nicole.ezer@mcgill.ca)
Asma Navasakulpong (navasakulpong@hotmail.com)
Kevin Schwartzman (kevin.schwartzman@mcgill.ca)
Linda Ofiara (linda.ofiara@mcgill.ca)
Anne Gonzalez (anne.gonzalez@mcgill.ca)

Version: 2 Date: 12 Oct 2017

Author’s response to reviews:

Minor editorial revisions has been completed for the manuscript entitled "Impact of Rapid Investigation Clinic on timeliness of lung cancer diagnosis and treatment".